Author's response to reviews

Title: Enhancing return-to-work in cancer patients, development of an intervention and design of a randomised controlled trial

Authors:

Sietske J Tamminga (s.j.tamminga@amc.uva.nl)
Angela GEM de Boer (a.g.deboer@amc.uva.nl)
Jos HAM Verbeek (j.h.verbeek@amc.uva.nl)
Taina Taskila (t.k.taskila@bham.ac.uk)
Monique HW Frings-Dresen (m.frings@amc.uva.nl)

Version: 3 Date: 1 July 2010

Author's response to reviews: see over
Authors’ response to comments editorial production team

The authors’ responses are in italic after each editorial production team comment

Title: Enhancing return-to-work in cancer patients, development of an intervention and design of a randomised controlled trial

Authors: Sietske J Tamminga, Angela GEM de Boer, Jos HAM Verbeek, Taina Taskila and Monique HW Frings-Dresen

Formatting changes to be made:

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

Each affiliation has been written as requested

Abstract: please ensure that the Abstract is present in the manuscript and matches exactly what is in the submission system, including all headings.

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings:
Background, Methods/Design, Discussion

The structure of the abstract has been reformatted after consulting with a technical editor from BMC Cancer as follows: Background, Methods/Design, Development and content of the intervention, Study design to evaluate the intervention, Discussion.

Manuscript: this manuscript does not have the correct structure. Please reformat it to use these headings:
Abstract; Background; Methods/Design; Discussion; Abbreviations (if any); Competing interests; Authors’ contributions; Authors¿ information (optional); Acknowledgements (optional); References

The structure of the manuscript has been reformatted after consulting with a technical editor from BMC Cancer as follows: Background, Methods/Design, Development of the intervention, Content of the intervention, Study design to evaluate the intervention, Discussion.

~

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
Email address: For the authors’ email address list, please use the following format as an example:
AB: abcd@institution.ac.uk
EF: efg@generic.co.uk
IJ: IJKL@corporation.com

Each email address has been written as requested.

Underlining: Please remove.

The underlining has been removed.

References: Please check if reference 15 'In Press' has now been published. If so, please provide full citation details. You can check here: http://www.ncbi.nlm.nih.gov/pubmed/

We have checked if reference 15 is published but it is not.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have checked our manuscript for typographical errors.

-----------------------------